Author's response to reviews

Title: Endocarditis caused by methicillin-susceptible Staphylococcus aureus with reduced susceptibility to vancomycin?: First case report in Argentina.

Authors:

Beatriz E Perazzi (hugodandrea@ciudad.com.ar)
Natalia Bello (nataliasbello@yahoo.com.ar)
Marta Mollerach (martamollerach@gmail.com)
Carlos A Vay (cavay@fibertel.com.ar)
María B Lasala (mblasala@fibertel.com.ar)
Angela M Famiglietti (famiglie@ffyb.uba.ar)

Version: 3 Date: 17 January 2011

Author's response to reviews: see over
Editorial Board
Journal of Medical Case Reports

MS: 9599549963715066 (Version 3)
Endocarditis caused by methicillin-susceptible Staphylococcus aureus with reduced susceptibility to vancomycin?: First case report in Argentina.

Beatriz E Perazzi, Natalia Bello, Marta Mollerach, Carlos A Vay, María B Lasala and Angela M Famiglietti

Dear members of the Editorial Board,

According to the suggestions of editors and reviewers, we have submitted our manuscript for further revision as regards language and style. The text was thoroughly checked by two certified translators working at a prestigious Argentine scientific Journal of Microbiology.

The changes in the revised manuscript are written in red.

We would appreciate your specific instructions for improving the quality and readability of our manuscript, should you find any other revisions or corrections to be addressed.

We are grateful for your valuable comments and suggestions as well as for your time and consideration.

Yours sincerely

Beatriz E. Perazzi, Professor Dr.
(on behalf of all authors)